

Received May 1, 2022, accepted May 15, 2022, date of publication May 20, 2022, date of current version May 27, 2022. Digital Object Identifier 10.1109/ACCESS.2022.3176614

Impact of Energy Imbalance on Financial Rewards in Peer-to-Peer Electricity Markets

TIMOTHY CAPPER[®], (Member, IEEE), JAISE KURIAKOSE[®], AND MARIA SHARMINA[®]

Tyndall Centre for Climate Change Research, School of Engineering, The University of Manchester, Manchester M13 9PL, U.K. Corresponding author: Timothy Capper (timothy.capper@manchester.ac.uk)

This work was supported by the Engineering and Physical Sciences Research Council (EPSRC) through the Power Networks Centre for Doctoral Training under Grant EP/L016141/1.

ABSTRACT Peer-to-peer electricity markets allow small producers and consumers of energy to trade directly. Energy imbalances, the difference between predicted and actual supply or demand for energy, may be higher in peer-to-peer markets than in traditional markets. High energy imbalances could increase the total cost of the electricity system, both inside and outside the peer-to-peer market, because the system operator must take expensive corrective actions at short notice. This paper examines the effect of imbalance charges on peer-to-peer electricity markets. A new symmetric imbalance charge mechanism is proposed which penalises market participants irrespective of the direction of their energy imbalance. The symmetric imbalance charge mechanism provides a financial incentive for peer-to-peer market participants to reduce their energy imbalances. For illustrative purposes, this paper uses a simulation to show that the current imbalance charge mechanism in Great Britain does not provide a financial incentive for peer-to-peer market participants to reduce their energy imbalances. The imbalance charge is close to zero under the current British imbalance charge mechanism when averaged over hundreds of settlement periods or more. When subject to the symmetric imbalance charge mechanism, the simulation shows that peer-to-peer market participants are given a strong financial incentive to reduce their energy imbalances. Finally, this paper discussed how the symmetric imbalance charge mechanism could be implemented only within the bounds of a peer-to-peer market, or within the whole electricity system.

INDEX TERMS Electricity market, energy imbalance, energy market, imbalance charge, imbalance price, peer-to-peer.

I. INTRODUCTION

Distributed generation is becoming a ubiquitous part of electricity systems. For example, the total installed capacity of distributed solar photovoltaic (PV) generation with an individual installed capacity of 5 MW or less per installation in Great Britain increased from 16 MW in January 2010 to 7,362 MW in January 2021 [1]. Feed-in tariff schemes are often the only way distributed generators can be compensated for exporting energy. For example, in Great Britain the Smart Export Guarantee scheme is the only revenue stream for distributed generators with a capacity less than 5 MW [2]. Peer-to-peer (P2P) electricity markets offer an alternative by allowing distributed generators to sell directly to consumers, without the need for an intermediary such as an energy supplier. Direct sale of electricity between distributed

The associate editor coordinating the review of this manuscript and approving it for publication was Hao Wang^(D).</sup>

generators and consumers represents a significant change in how electricity markets function. The current codes and standards which govern electricity markets will not necessarily be appropriate for P2P markets. This paper proposes a symmetric imbalance charge mechanism which provides P2P electricity market participants with a financial incentive to minimise their energy imbalances. To demonstrate the effectiveness of the symmetric imbalance charge mechanism, it is contrasted with the current British imbalance charge mechanism.

A. ENERGY IMBALANCES AND IMBALANCE CHARGES

In liberalised electricity markets, such as those found in Britain and much of Europe, competitive wholesale market trading is followed by a monopolistic balancing mechanism run by a system operator to ensure energy supply and demand are matched [3]–[5]. Trading in the wholesale market is based on market participants' predictions of their supply and demand for energy. Errors in these forecasts, known as energy imbalances, are corrected by the system operator in the balancing mechanism.

Imbalance charges compensate the system operator for their costs in the balancing mechanism [3]. Imbalance charges are paid by market participants to the system operator in proportion to their energy imbalance.

Different regulatory regimes place balancing responsibility on different parties. In Britain and much of Europe, balancing responsibility is placed on individual market participants, by requiring them to balance their position in wholesale markets as far as possible [3]–[5]. In other regulatory regimes the system operator manages the cumulative energy imbalances of all market participants, which will be lower than the sum of the magnitude of individual imbalances, e.g. the PJM region of the USA [6]. It is debatable which balancing mechanism design is most efficient. In this paper we assume that we are in a regulatory regime that places balancing responsibility on market participants. That is the case for Great Britain, used in our comparative case study. Section II provides a full description of imbalance charges in Great Britain.

B. ENERGY IMBALANCES IN PEER-TO-PEER ELECTRICITY MARKETS

Peer-to-peer electricity markets are likely to exist, at least initially, as sub-elements of traditional electricity markets. The most developed P2P market models are relatively small markets that only contain small energy producers and consumers [7]. Peer-to-peer markets in this form cannot supplant traditional markets. Therefore, it is likely that P2P markets will be subject to the codes that govern traditional electricity markets, including those concerning balancing responsibility.

Existing P2P market literature focuses on optimising the internal working of P2P markets [8]. While this work is valuable, if P2P markets are to exist the internal design of the markets must account for the regulations to which they are subject. The regulations must also account for P2P markets [9]. Literature which compares the profitability of traditional and P2P electricity markets finds that participants are more profitable in P2P markets [10]–[12]. However, most of the literature examining P2P electricity market design assumes participants can accurately predict their supply and demand for energy [12]–[17]. In reality this is unlikely to be the case [18], [19]. Therefore, P2P market participants will be subject to imbalance charges. When taking the retail market price and feed-in tariff, market participants are not subject to imbalance charges. The additional cost of imbalance charges is neglected by the cost models in the current literature. This means that profitability in real-world P2P markets is likely to be lower than modelled in the current literature.

Pilot projects of P2P markets tend to be run in regulatory sandboxes [20]. These sandboxes exempt the pilot projects from much of the regulation of electricity markets, including imbalance charges [21]. Therefore, pilot projects have not provided evidence of the energy imbalance which could be expected from P2P markets, or of how imbalance charges would affect P2P markets. A recent set of P2P market literature has examined how energy imbalances in P2P market could be minimised through optimal scheduling of flexible resources [11], [22]–[29], or the use of real-time markets [30]–[34]. However, these markets do not completely eliminate energy imbalances and therefore an imbalance charge mechanism would still be required. The symmetric imbalance charge proposed in this paper would complement P2P market designs which minimise energy imbalances by providing an external financial incentive to market participants who reduce their energy imbalances.

C. OUR CONTRIBUTION

Many current imbalance charge mechanisms, including the one in Britain, are audit and fines based [35]. These imbalance charge mechanisms do not provide a direct financial incentive for market participants to minimise their energy imbalances, but fine participants who do not [36]. While this audit and fines system is workable for the relatively small number of market participants which exist in current wholesale energy markets, even relatively small P2P markets could dramatically increase the number of balance responsible parties [37]. This dramatic increase in the number of balance responsible parties which could accompany the introduction of P2P energy markets necessitates a rethink of how market participants are incentivised to minimise their energy imbalances.

In this paper we propose a symmetric imbalance charge mechanism which provides a financial incentive for market participants to minimise their energy imbalances. This financial incentive removes the need for an audit and fines based enforcement system and is therefore more appropriate for the large numbers of balance responsible parties which can be expected with the advent of P2P energy markets. The symmetric imbalance charge mechanism proposed in this paper differs from the current British imbalance charge mechanism [3], and others discussed in recent academic literature [38] in that market participants are charged the imbalance charge irrespective of the direction of their energy imbalance. This provides a strong financial incentive for market participants to minimise their energy imbalances. A full description of the symmetric imbalance charge mechanism is provided in Section III.

We present a Monte Carlo simulation of households with PV acting in a P2P energy market to demonstrate the contrast in financial incentives between our proposed symmetric imbalance charge mechanism, and the current British imbalance charge mechanism. This simulation shows that the imbalance charge is a significant proportion of profit in P2P markets when energy imbalances are high, but falls significantly when energy imbalances are reduced. This significant change in imbalance charges proportionate to energy imbalances shows that P2P market models must consider energy imbalances in their market costs.

Previous literature has shown that if P2P market participants were subject to imbalance charge mechanisms such as the one which exists in Britain today, they could game the system to financially benefit from the imbalance price [39]–[42], described as malicious behaviour throughout the remainder of this paper. We model two types of malicious behaviour in our Monte Carlo simulation and demonstrate that market participants cannot benefit by gaming our proposed symmetric imbalance charge mechanism. This inability to game the system is important as it further removes the need for an audit based enforcement system of balancing responsibility.

The rest of the paper is structured as follows. Section II provides a mathematical derivation of imbalance charges as they currently exist in Great Britain. Section III provides a mathematical derivation of the symmetric imbalance charge mechanism. Section IV provides detail of the methodology and data used. Section V provides the results of our model and analysis of the results. Finally, Section VI provides concluding remarks. Additional supporting results data are provided in Appendix A.

II. IMBALANCE CHARGES IN GREAT BRITAIN

In Great Britain, imbalance charges are governed by the Balancing and Settlement Code (BSC) [3]. BSC parties are responsible for predicting their supply and demand for energy, and putting contracts in place to buy or sell excess energy. After gate closure, the system operator takes over responsibility for balancing supply and demand for energy.

The system operator, National Grid ESO in Britain, uses the balancing mechanism to keep the supply and demand for energy in balance during the settlement period. Generators and demand response operators acting in the balancing market submit bids and offers to increase or reduce their supply or demand for energy during a settlement period. Bids to increase supply or reduce demand have a positive price, i.e. the system operator must pay. Offers to reduce supply or increase demand have a negative price, i.e. the system operator will be paid. Bids and offers are selected by the system operator through merit ordering. The system operator's energy balancing costs are covered by an imbalance price paid by BSC parties proportionate to their energy imbalance.

The imbalance price, λ_{imb}^k , for a settlement period, k, is set at the marginal cost of the balancing mechanism [35]. The imbalance price is therefore the price of the most expensive balancing action, bm, from the set of balancing actions during that settlement period, BM. This incentivises BSC parties to take all actions to reduce energy imbalances which cost less than it would cost the system operator. This idea, that the party who can take the balancing action for the lowest cost should take it, minimises the total system cost.

$$\lambda_{imb}^{k} = \max_{bm \in BM} \left\{ \lambda_{bm}^{k} \right\} \tag{1}$$

The imbalance price is normally positive, although it can be negative [43]. If all balancing mechanism actions the system operator has ordered during a settlement periods are reductions in supply or increases in demand then the cost of the marginal balancing mechanism action might be negative. Market participants responsible for energy imbalances (BSC parties) must submit the amount of energy they have bought or sold for a settlement period to the system operator before the beginning of that settlement period, a point called gate closure [35]. This submission by a market participant, *i*, is the final notification volume, $E_{fnv}^{i,k}$. The energy imbalance, $E_{inb}^{i,k}$, is the difference between the final notification volume, $E_{fnv}^{i,k}$, and net exported energy, $E_{exp}^{i,k}$ are both positive for export and negative for import. The energy imbalance, $E_{imb}^{i,k}$, is positive when the market participant exports too little or import too little energy, and negative if they export too much or import too little energy.

$$E_{imb}^{i,k} = E_{fnv}^{i,k} - E_{exp}^{i,k}$$
(2)

The imbalance charge incurred by a BSC party during a settlement period, $C_{imb}^{i,k}$, is the energy imbalance of that BSC party during the settlement period, $E_{imb}^{i,k}$, multiplied by the imbalance price, λ_{imb}^k . Figure 1 shows a timeline of energy trading in Great Britain.

$$C_{imb}^{i,k} = E_{imb}^{i,k} \lambda_{imb}^k \tag{3}$$

The imbalance charge, $C_{imb}^{i,k}$, is positive when a BSC party's energy imbalance, $E_{imb}^{i,k}$, is in the same direction as the imbalance of the whole grid. If the whole grid has a deficit of energy the system operator must pay generators to increase their supply of energy, or pay loads to reduce their demand. Therefore the imbalance price, λ_{imb}^k , will be positive. BSC parties who have a deficit of energy during a settlement period have a positive energy imbalance, $E_{imb}^{i,k}$. These BSC parties will receive a positive imbalance charge. Conversely, if a BSC party has an imbalance in the opposite direction to the whole grid, they will have a negative imbalance charge, paid by the system operator to the BSC party.

The system operator's revenue, R_{SO}^k , is based on the net energy imbalance. There is a single imbalance price for each settlement period. Any BSC parties who have an energy imbalance in the opposite direction to the whole grid will be paid the imbalance charge. Therefore, the system operator will only receive the imbalance charge proportionate to the net imbalance in the system. So, from equations 1 and 3 the system operator's revenue is the sum of the imbalance charges over the set of market participants, *I*.

$$R_{SO}^{k} = \sum_{i \in I} C_{imb}^{i,k} \tag{4}$$

$$=\sum_{i\in I} E_{imb}^{i,k} \cdot \max_{bm\in BM} \left\{ \lambda_{bm}^k \right\}$$
(5)

The principle of paying BSC parties who have an imbalance in the opposite direction to the whole grid is based on the idea that their imbalance is beneficial and therefore they should be compensated. Although these BSC parties have provided a service to the grid, they did not do so intentionally. Balancing market participants who adjust their supply and



FIGURE 1. Timeline of energy trading in Great Britain.

demand at the request of the system operator to balance the grid are compensated separately.

III. SYMMETRIC IMBALANCE CHARGE MECHANISM

To encourage P2P electricity trading, we propose a new method for calculating imbalance charges. This symmetric imbalance charge penalises all energy imbalances, irrespective of whether they are in the same direction as the imbalance of the whole grid. Where the purpose of many current imbalance charge mechanisms is to recuperate energy balancing costs, the symmetric imbalance charge also aims to provide an incentive for parties to reduce their level of imbalance. This financial incentive is provided through the imbalance price, without the need for a separate levy. We specify three criteria which any new imbalance charge mechanism must meet to be compatible with P2P electricity markets:

- The total imbalance charge must equal or exceed the energy balancing costs of the system operator: The imbalance charge is the only mechanism by which the system operator is compensated for their energy balancing service. Therefore it is important that any change to the imbalance charge maintains the system operator's revenue.
- 2) The imbalance charge must provide an incentive for P2P market participants to reduce their energy imbalance: Implementing measures to reduce energy imbalance is expensive. For example P2P market participants could install energy storage or flexible loads. Therefore a financial incentive must be provided.
- 3) The imbalance charge must be low enough that a P2P market is beneficial over a traditional electricity market: If imbalance charges are high enough to eliminate profits in P2P electricity markets, then participants will revert to traditional energy markets.

The P2P market price is effectively bounded by the retail market price, charged by energy suppliers, and the feed-in tariff. If the P2P market price is above the retail market price for a sustained period, market participants will switch to purchasing energy from the retail market. If the P2P market price falls below the feed-in tariff, market participants will switch to selling energy through the feed-in tariff. Market participants must be able to agree to a price within this range that is beneficial to both the buyer and seller.

If the imbalance charge is larger than the difference between the retail market price and the feed-in tariff, it will prevent P2P markets from being possible. For simple P2P markets which only trade energy, the P2P market may still function with a single price. However, P2P markets which use dynamic pricing to balance supply and demand of energy form the bulk of the current literature [13]–[16], [44]. Markets with dynamic pricing may require a wider band of allowable market prices. The size of imbalance charge which would prevent P2P markets from functioning therefore depends on the aim and design of the market.

Under the symmetric imbalance charge mechanism, market participants would be charged the imbalance price, λ_{imb}^k , irrespective of whether their energy imbalance is in the same direction as the whole grid. The imbalance price has a lower bound of the highest market price for that settlement period. This combination of always charging the imbalance charge, and the imbalance price always being equal to or above the highest market price prevents market participants gaming the imbalance system for their benefit, as discussed further in Section III-B.

$$C_{imb}^{i,k} = \begin{cases} \left| E_{imb}^{i,k} \right| \cdot \lambda_{imb}^{k}, & \text{if } \lambda_{imb}^{k} \ge \max \left\{ \lambda_{P2P}^{i,k} \right\} \\ \left| E_{imb}^{i,k} \right| \cdot \max \left\{ \lambda_{P2P}^{k} \right\}, & \text{if } \lambda_{imb}^{k} < \max \left\{ \lambda_{P2P}^{i,k} \right\} \end{cases}$$
(6)

Under current imbalance charge mechanisms, the system operator's revenue is based on the net energy imbalance. Under the symmetric imbalance charge mechanism the system operator's revenue is based on gross imbalance. Therefore the imbalance price, λ_{imb} , can be reduced while maintaining the system operator's revenue. Section II provides a derivation of the system operator's revenue under the current British imbalance charge mechanism. The revenue of the system operator under the symmetric imbalance charge mechanism is the sum of the imbalance charges. Note for

equations 7 through 12 only we use subscript 'sym' when refer to the symmetric imbalance charge mechanism and 'cur' when referring to the current British imbalance charge mechanism.

$$R_{SO,sym}^{k} = \sum_{i \in I} \left(\left| E_{imb}^{i,k} \right| \cdot \lambda_{imb,sym}^{k} \right)$$
(7)

The aim is to calculate the imbalance price at which the system operator's revenue is maintained. Therefore equating the system operator's revenue under the current imbalance charge mechanism from equation 5 with the system operator's revenue under the symmetric imbalance charge mechanism from equation 7...

$$\sum_{i \in I} E_{imb}^{i,k} \cdot \max_{bm \in BM} \left\{ \lambda_{bm}^k \right\} = \sum_{i \in I} \left(\left| E_{imb}^{i,k} \right| \cdot \lambda_{imb,sym}^k \right)$$
(8)

Substituting in equation 1...

$$\sum_{i \in I} E_{imb}^{i,k} \cdot \lambda_{imb,cur}^{k} = \sum_{i \in I} \left| E_{imb}^{i,k} \right| \cdot \lambda_{imb,sym}^{k}$$
(9)

Defining the ratio of imbalance price under the symmetric mechanism to the imbalance price under the current British imbalance charge mechanism as α ...

$$\lambda_{imb,sym}^{k} = \alpha \cdot \lambda_{imb,cur}^{k} \tag{10}$$

Therefore, substituting equation 10 into 9...

$$\sum_{i \in I} E_{imb}^{i,k} \cdot \lambda_{imb,cur}^k = \sum_{i \in I} \left| E_{imb}^{i,k} \right| \cdot \alpha \cdot \lambda_{imb,cur}^k$$
(11)

$$\alpha = \frac{\sum_{i \in I} E_{imb}^{i,k}}{\sum_{i \in I} \left| E_{imb}^{i,k} \right|}$$
(12)

Since $\sum_{i \in I} \left| E_{imb}^{i,k} \right| \geq \sum_{i \in I} E_{imb}^{i,k}$, the imbalance price, λ_{imb}^k , under the symmetric mechanism will always be less than or equal to the price under the current British imbalance charge mechanism. These calculations assume that the bids and offers in the balancing market remain the same under both mechanisms.

A. NET BENEFIT OF A PEER-TO-PEER MARKET

To determine if market participants are better off in a P2P market or a traditional energy market, the profit of a market participant is derived. The profit of a P2P market participant, π_{P2P}^{i} , is the net energy exported during a settlement period, $E_{exp}^{i,k}$, multiplied by the price the energy is sold for in that period, λ_{P2P}^{k} , minus any imbalance charges incurred during that period, $C_{imb}^{i,k}$, summed for the set of settlement period in question, K.

$$\pi_{P2P}^{i} = \sum_{k \in K} (E_{exp}^{i,k} \lambda_{P2P}^{k} - C_{imb}^{i,k})$$
(13)

The profit of a participant in a traditional electricity market, π_{trad}^{i} , is the net exported energy, $E_{exp}^{i,k}$, multiplied by the feedin tariff, λ_{FIT} , if the net exported energy is positive. The profit is the net exported energy, $E_{exp}^{i,k}$, multiplied by the retail market price, λ_{trad} , if the net exported energy is negative (i.e. a net import of energy). The net exported energy, $E_{exp}^{i,k}$, is the energy generation minus demand during that settlement period.

$$\pi_{trad}^{i} = \begin{cases} E_{exp}^{i,k} \lambda_{FIT}, & \text{if } E_{exp}^{i,k} > 0\\ E_{exp}^{i,k} \lambda_{trad}, & \text{if } E_{exp}^{i,k} < 0 \end{cases}$$
(14)

B. MALICIOUS BEHAVIOUR

The method of calculating imbalance charges in the British BSC (see Section II) allows market participants to intentionally adjust their energy market position to benefit from imbalance charges. These actions are not beneficial to the market participant under the proposed symmetric imbalance charge mechanism. This section introduces two types of malicious behaviour and shows that they are beneficial to the market participant under the current British imbalance charge mechanism, but not under the symmetric imbalance charge mechanism.

1) PURCHASING MORE ENERGY THAN REQUIRED

The imbalance price is normally above the P2P market price, $\lambda_{imb}^k > \lambda_{P2P}^k$, based on historic data [43]. When under the current British imbalance charge mechanism participants are paid the imbalance price for excess energy. Therefore, if the market participant intentionally creates a long position in the market by purchasing more energy than they predict they will need, or not selling energy they predict they will generate they will be paid the imbalance price for this energy during settlement. Since the imbalance price is normally above the market price they will, on average, financially benefit from this action.

Acting maliciously in this manner is low effort for the market participants. They do not need to make any predictions of the market prices in order to benefit from this strategy. However, if the market participants are able to predict the market price and imbalance price, then they can gain greater financial benefit by using this information.

The symmetric imbalance charge mechanism prevents market participants from acting maliciously in this manner. Under the symmetric imbalance charge mechanism participants are penalised irrespective of the direction of their imbalance. Therefore, participants will be penalised for a long market position.

2) ADJUSTING POSITION BASED ON IMBALANCE PRICE PREDICTION

If the imbalance price is below the market price during a settlement period, $\lambda_{imb}^k < \lambda_{P2P}^k$, market participants can benefit under the current British imbalance charge mechanism by creating a short position. Market participants could sell more energy than they anticipate generating, or buy less than they anticipate demanding in the P2P market. The market participant will be forced to purchase this deficit in energy at the imbalance price during settlement. However, if the

imbalance price is below the market price, then they will be better off.

Acting maliciously in this manner is higher effort than the manner described in Section III-B1. The imbalance price is normally above the market price. Market participants must be able to predict the settlement periods in which the imbalance price will fall below the market price in order to profitably act on this strategy. If they can accurately predict the imbalance price, they could combine both strategies depending on the relative level of imbalance and market prices.

The symmetric imbalance charge prevents this behaviour by putting a lower bound on the imbalance price at the P2P market price. If the imbalance price never falls below the market price then acting maliciously in this manner is never financially beneficial.

IV. METHODS AND DATA

To demonstrate the symmetric imbalance charge mechanism we present a Monte Carlo simulation of households with PV acting in a P2P electricity market. We subject the households to the symmetric imbalance charge mechanism and for comparison also to the current British imbalance charge mechanism. In this section we present the datasets we have used, followed by the method of analysis and finally the key assumptions.

A. INPUT DATA

We have used three datasets in our modelling, covering household demand, rooftop PV generation and imbalance prices.

1) HOUSEHOLD DEMAND

The UK Power Networks' SmartMeter Energy Consumption Data in London Households dataset [45] contains demand data from 5,567 households in London, UK. The data was collected between November 2011 and February 2014. The length of data collection varies for each household. We extracted 1,192 year long datasets which do not contain any missing data points. The datasets contain the energy consumed during each half-hour settlement period for one year. The maximum demand of the households varied from 0.022 kW to 18.282 kW across the different datasets. During each model run one of the 1,192 demand datasets was randomly selected.

2) ROOFTOP PV GENERATION

The UK Power Networks' Photovoltaic (PV) Solar Panel Energy Generation dataset [46] contains data from six rooftop PV installation in South East England. The data was collected between July 2013 and November 2014. The data from June to November 2014 is provided in 10 minute intervals. The data prior to June 2014 is provided as hourly minima and maxima. We have created six year-long datasets with a halfhour temporal resolution. The datasets contain energy generated by the PV installation during each half-hour settlement period for one year. Where available we have summed the 10 minute resolution data into 30 minute periods. For the periods where 10 minute resolution data was not available we have used the hourly data. We have taken a random value from a uniform distribution bounded by the minimum and maximum recorded value for that hour. The installed capacity of the PV installations ranged from 0.444 kW to 3.965 kW.

We used the European Commission's PV-GIS system [47] to calculate the theoretical difference in generation between a PV installation in the most Northerly and Southerly location. The difference is less than 4%, much smaller than the difference due to the variation in installed capacity between the installations. The largest PV installation in the dataset is almost nine times the size of the smallest. Therefore, it is reasonable to use data from the different PV installation in the same P2P market in the model. During each model run one of the 6 generation datasets was randomly selected.

3) IMBALANCE PRICES

We have used imbalance prices from 2019 and 2020 in Great Britain. The data was collected from the Elexon Data Portal [43]. The two datasets contain the British imbalance price for each half hour settlement period in the year. During each model run one of the 2 imbalance price datasets was randomly selected. Imbalance prices from before 2019 were not used because the way the imbalance price was calculated in Great Britain changed at the end of 2018 [48].

4) PEER-TO-PEER MARKET PRICE

The P2P market price is bounded by the retail market price and the feed-in tariff. The mean standard variable tariff in Great Britain, 14.4 p/kWh [49], is the upper bound of the P2P market price. The highest feed-in tariff (Smart Export Guarantee) price, 5.6 p/kWh [50], is the lower bound of the P2P market price. These values have been chosen because they were the values published. During each model run either 14.4 p/kWh or 5.6 p/kWh is randomly selected as the P2P market price. Hence, the prices tested in this model are the maximum and minimum possible P2P market prices.

B. MODEL

We use a Monte Carlo simulation [51], [52] to examine the effect of imbalance charge design on the profitability of a household in a P2P electricity market. The model is of a single household with load and a rooftop PV installation trading in a P2P electricity market. The load and generation profiles are randomly allocated at the beginning of the model run. See Sections IV-A1 and IV-A2 for more details about the datasets. We assume that the household is responsible for their energy imbalances. Therefore, the household will be penalised for any energy imbalances they incur.

The aim of this model is to identify the range of imbalance charges and profits which are feasible under both the symmetric imbalance charge mechanism and the current British imbalance charge mechanism for comparison. The purpose of this model is to test the effectiveness of the symmetric imbalance charge mechanism in providing a financial incentive to minimise energy imbalances. This model does not optimise the behaviour of the market participants subject to the imbalance charges. Optimal scheduling of resources to minimise energy imbalances is the subject of other works [11], [22]–[34].

The household acts in a P2P market which is sufficiently large that it can accommodate all their supply and demand for energy during every settlement period. It is assumed that all excess supply from the household is sold at the market price and that all excess demand is purchased at the market price. The rest of the P2P market is not explicitly modelled. This assumption is discussed further in Section IV-C.

The model has a settlement period of 30 minutes and gate closure is one hour before the start of the settlement period. The length of settlement period and gate closure are those currently used in Great Britain. At gate closure the household estimates their supply or demand of energy for the settlement period. The P2P market is an hour ahead futures market to conform with the gate closure requirement. The household's estimated supply or demand of energy is bought or sold in the P2P market. Any difference between the estimated supply or demand and the actual supply or demand for energy during the settlement period is an energy imbalance. The household is penalised for energy imbalances according to the imbalance charge mechanism.

The household estimates their supply and demand for energy using a simple moving average with a backward window of one. A simple moving average is a computationally simple estimation method that would be easy to implement. It is likely that more sophisticated and accurate estimation methods would be used in a real P2P market. The aim of using a simple moving average in this model is that it is the worst estimate of supply and demand that a household could credibly make.

Simple moving average backward windows of one, 2, 3, 4, 5, 7 and 10 were tested. The backward window determines how many previous periods the simple moving average averages over. A backward window of one gives the best estimate of energy supply and demand using a simple moving average.

The μ_{pred} parameter replicates improvements in the prediction of energy supply and demand. We test a uniform distribution of μ_{pred} between zero and 1 in steps of 0.1. A μ_{pred} value of one is equivalent to the simple moving average estimate. Smaller values of μ_{pred} reduce energy imbalances, mimicking installation of storage or flexible load which can dynamically adjust its demand or supply to meet the prediction, or trading within the P2P market to net-off imbalances.

The household is subject to one of two imbalance charge mechanisms: either the symmetric imbalance charge mechanism (Equation 6), or the current British imbalance charge mechanism (Equation 3). When subject to the symmetric imbalance charge mechanism, the imbalance price coefficient, μ_{imb} , is selected from a uniform distribution between zero and 1 in steps of 0.1. The imbalance price coefficient reduces the imbalance price from today's level, mimicking

a reduction to maintain the system operators revenue (see Equation 12). The imbalance charge mechanism and imbalance price coefficient are constant during each model run.

The household can either act honestly or maliciously in the model (see Section III-B). If the household acts honestly, they buy or sell their estimated supply or demand for energy in the futures market. If the household acts maliciously, they adjust their futures market position in an attempt to financially benefit from the imbalance charge. Two types of malicious behaviour are modelled.

Under the first type of malicious behaviour (Section III-B1), a household reduces their estimated export of energy by 50% if it is positive, and increase it by 50% if it is negative. These changes to the final notification volume make households acting maliciously more likely to have a negative energy imbalance.

Under the second type of malicious behaviour (Section III-B2), the household predicts the relative levels of market price and imbalance price. If the imbalance price is above the market price the household reduces their estimated export of energy by 50% if it is positive, and increases it by 50% if it is negative. If the imbalance price is below the market price the household increases their estimated export of energy by 50% if it is positive, and reduces it by 50% if it is negative. In the model there is a 60% chance a household will decide to act honestly, and a 20% chance they will choose each of the malicious behaviours.

The choice to change final notification volumes by 50% in these models of malicious behaviour has been selected by the authors for illustrative purposes and does not constitute a threshold value. The same effect can be observed to a greater or lesser extent at any percentage change. When designing this model a variety of percentage changes were tried and the value of 50% has been presented in this paper because the effect is visible on small graphs. The actual percentage changes used by market participants acting maliciously in reality would be a trade-off between maximising their gain while trying to avoid detection.

For this Monte Carlo simulation, 10,000 model runs were performed. Each run was 10,000 settlement periods long, equivalent to 208.3 days. Algorithm 1 shows pseudo-code of the model.

C. ASSUMPTIONS

In the course of building the model, four important assumptions have been made.

1) EACH HOUSEHOLD IS LIABLE FOR THEIR

ENERGY IMBALANCES

This means that each household must notify the system operator of their supply or demand for energy an hour before the settlement period. They must accept any imbalance charges due to deviation from that final notification volume. In traditional energy markets, energy suppliers are responsible for energy imbalances on behalf of households. There are two ways individual households might become liable for their energy imbalances in P2P markets. Firstly, every household in a P2P market could become a BSC party. This would be a significant change from the current system, but has been suggested in some literature [38], [53]. Secondly, a market operator could act as the BSC party for the whole P2P market, but directly pass the imbalance charges onto the market participants. This arrangement may be possible under the Virtual Power Plant provision in the BSC [3]. The results of this model are not affected by the method in which imbalance charges are passed to households.

Algorithm 1 Imbalance Charge Model Pseudo Code

Initialise model

select randomly Demand profile \in set of 1192 Generation profile \in set of 6 Imbalance price profile \in set of 2 $\lambda_{P2P} \in \{5.6, 14.4\} p/kWh$ $\mu_{pred} \in 0 \le \mu_{pred} \le 1$, step 0.1 $\mu_{imb} \in 0 \le \mu_{imb} \le 1.1$, step 0.1 $\mu_{mal} \in \{1, 1, 1, 2, 3\}$

Run model

for k = 1 to 10000

Calculate exported energy.

$$E_{exp}^k = E_{gen}^k - E_{dem}^k$$

Calculate futures market position. The predicted energy export is the energy exported in the settlement period immediately prior to gate closure with the error reduced depending on μ_{nred} .

the error reduced depending on μ_{pred} . $E_{pred}^{k} = E_{exp}^{k} + \mu_{pred}(E_{exp}^{k-3} - E_{exp}^{k})$ **if** $\mu_{mal} = 1$

If the household is acting honestly then they will trade their predicted export in the P2P market.

 $E_{P2P}^{k} = E_{pred}^{k}$

else if
$$\mu_{mal} = 1$$

If the household is acting maliciously by just increasing their predicted export then:

if
$$E_{pred}^k > 0$$

 $E_{P2P}^k = E_{pred}^k \cdot 1.5$
else
 $E_{P2P}^k = E_{pred}^k \cdot 0.5$

else

If the household is acting maliciously by adjusting the market position based on a prediction of the relative levels of the market and imbalance prices then:

$$\begin{aligned} \text{if } \lambda_{imb}^{k} > \lambda_{P2P}^{k} \& E_{pred}^{k} > 0 \\ E_{P2P}^{k} = E_{pred}^{k} \cdot 1.5 \\ \text{else if } \lambda_{imb}^{k} > \lambda_{P2P}^{k} \& E_{pred}^{k} < 0 \\ E_{P2P}^{k} = E_{pred}^{k} \cdot 0.5 \end{aligned}$$

else if $\lambda_{imb}^k < \lambda_{P2P}^k$ & $E_{pred}^k > 0$ $E_{P2P}^k = E_{pred}^k \cdot 0.5$ else if $\lambda_{imb}^k < \lambda_{P2P}^k$ & $E_{pred}^k < 0$ $E_{P2P}^k = E_{pred}^k \cdot 1.5$ else $E_{P2P}^{k} = E_{pred}^{k}$ Calculate the profit from the traditional market. if $E_{exp}^k > 0$ $\pi_{trad}^{k} = E_{exp}^{k} \lambda_{FIT}$ else $\pi^k_{trad} = E^k_{exp} \lambda_{trad}$ Calculate the energy imbalance. $E_{imb}^k = E_{P2P}^k - E_{exp}^k$ If $\mu_{imb} = 1.1$ the current British imbalance charge mechanism is used. If $0 \le \mu_{imb} \le 1$ the symmetric imbalance charge mechanism is used. if $\mu_{imb} = 1.1$ $C_{imb}^k = E_{imb}^k \lambda_{imb}^k$ else $\text{if }\lambda_{imb}^k < \lambda_{P2P}$ If the imbalance price is lower than the P2P market price the imbalance price will be set at the market price. $\begin{aligned} \lambda_{imb}^{k} &= \lambda_{P2P} \\ C_{imb}^{k} &= |E_{imb}^{k}| \; \lambda_{imb}^{k} \; \mu_{imb} \end{aligned}$ Calculate the additional profit from the P2P mar- $R_{P2P}^k = E_{P2P}^k \lambda_{P2P}$ $\pi_{P2P}^{k} = R_{P2P}^{k} - C_{imb}^{k}$ P2P market additional profit = $\pi_{P2P}^{k} - \pi_{trad}^{k}$

2) THE HOUSEHOLD CAN BUY OR SELL ALL THEIR EXCESS ENERGY IN THE P2P MARKET

The model assumes that there is sufficient demand in the P2P market to purchase all excess energy the household has, and there is sufficient supply to meet all the demand the household has. This means that the household can always sell all their excess energy and they can always buy all the energy they need to meet a deficit. In a real P2P market it is unlikely that supply and demand for energy would perfectly balance during every settlement period as assumed in this model.

In the current literature, some P2P market models aim to perfectly balance energy supply and demand within the market [11], [54], [55], as assumed in this paper, but many assume that the traditional energy market acts as a supplier of last resort [13]–[16]. In the case that the household purchases energy from the retail market, they are likely to pay the retail market price, which is higher than the P2P market price. In the case that the household cannot sell their energy in the P2P market, they may be able to sell it at the feed-in tariff which is lower than the P2P market price, or they may not be able to sell it at all. It is therefore possible that the revenue of a household in a real P2P market will be lower than in the model in this paper.

3) THE ENERGY IMBALANCE OF THE P2P MARKET DOES NOT AFFECT THE IMBALANCE PRICE

The model assumes there is no correlation between the household's energy imbalance and the imbalance price. If P2P markets form only a small proportion of the total supply and demand in the electricity system, there will be little or no correlation between the energy imbalance of the household in the P2P market and the imbalance price. If the network energy imbalance is correlated with the P2P market energy imbalance, the imbalance charge will become correlated with the energy imbalance of the household.

Errors in widely used forecasts could also lead to energy imbalances correlating with the imbalance price. If a large proportion of a country's energy market participants are relying on the same weather forecast, which is inaccurate, then they are likely to have energy imbalances in the same direction. This error could lead to a correlation between a national imbalance price and energy imbalance in a small P2P market. We do not consider this correlation in this paper.

THE P2P MARKET PRICE IS NOT ADJUSTED FOR NON-ENERGY COSTS

The P2P market price in the model is based on the current electricity market prices in Great Britain. These prices include system operator and network operator fees, including the imbalance charge. In Britain network charges account for 22% of an average customer bill, and operating costs account for 17% [56]. Operating costs include imbalance charges and non-energy balancing costs such as the cost of reserve and geographical balancing. In reality the lower bound of the P2P market price may be slightly higher and the upper bound may be slightly lower. This is because the imbalance charges are calculated separately in the P2P market.

V. RESULTS AND ANALYSIS

The results show that household revenue is mostly higher in the P2P market than the traditional market. Only 0.3% of households not acting maliciously (18 of 5,928) had a higher revenue in the traditional market. However, the imbalance charges experienced by households can dramatically affect the profits households make in the P2P market. Even though 99.7% of households not acting maliciously (5,910 of 5,928) had a higher revenue in the P2P market, only 51.6% of households (3,059 of 5,928) had a higher profit.

In this section the results from the Monte Carlo simulation are analysed to explain why imbalance charges have such a significant effect on the profits in P2P markets. Section V-A analyses the revenues of households. Section V-B analyses the energy imbalances of the households. Section V-C analyses the imbalance charges of households subject to the symmetric imbalance charge mechanism and Section V-D analyses the profits. Sections V-E and V-F analyse the imbalances charges and profits of households subject to the current British imbalance charge mechanism. Sections V-B - V-F discuss the results of households which act honestly in the market. Section V-G analyses whether households can adjust their supply and demand predictions to improve their profits. Section V-H discusses the economic efficiency of the symmetric imbalance charge mechanism. Section V-I discusses how the symmetric imbalance charge mechanism might be implemented. Finally, Section V-J discusses methods by which P2P market participants might reduce their energy imbalances in response to the imbalance charge.

A. HOUSEHOLD REVENUES

The total revenue of households in the P2P market at the end of the model run forms a bi-modal distribution with a large negative skew. The highest peak of the distributions is at a revenue of 118.37 GBP/year and the secondary peak is at -57.87 GBP/year. The maximum and minimum revenues at the end of the model runs were 717.96 GBP/year and -5,028.32 GBP/year respectively. The shape of the revenue distribution is determined by the generation and demand of the household.

The bi-modal nature of the distribution is caused by the generation datasets. Of the six PV generation datasets used in the model, four have a mean generation between 0.211 kWh and 0.287 kWh per settlement period, while the other two have mean generations of 0.025 kWh and 0.050 kWh per settlement period. Households which used one of the higher generation profiles are more likely to have higher net energy exports, and therefore high revenues. These households with higher generations profiles form the peak at a revenue of 118.37 GBP/year. The households using one of the lower two generation profiles had lower net energy exports and so formed the lower peak at a revenue of -57.87 GBP/year. The bi-modal nature of the distribution would not exist in reality but is caused by the small number of PV generation profiles used in the model.

The large negative skew of the revenue distribution is caused by the demand datasets. The mean of the distribution of mean energy demand was 0.224 kWh per settlement period. However, the distribution has a large positive skew. The minimum mean energy demand is 0.001 kWh per settlement period. The maximum mean energy demand is 2.043 kWh per settlement period. This large positive skew is caused by a small number of demand datasets which have very large energy demands compared to the rest. Only two of the 1,192 demand datasets have a mean energy demand of over 1.5 kWh per settlement period, only 7 have a mean energy demand of over 1 kWh per settlement period. House-holds with the large demand datasets have low net exports and therefore low revenues compared to other households.

However, the absolute revenue of the households in the P2P market matters less than relative revenue compared to the traditional market. Of the households not acting maliciously by intentionally adjusting their energy prediction, only 18 of 5,928 households had a lower revenue in the P2P market than the traditional market. The reason most households have

a higher revenue in the P2P market is that there is no spread between buy and sell prices in the market. Households in the P2P market are either paid more for exporting electricity, or pay less to import electricity than in the traditional market. The reason a small number households were better off in the traditional market is that their actual energy export was significantly greater than their predicted energy export.

The revenue of households in the P2P market does not take account of imbalance charges caused by their energy imbalances. The energy imbalances have a significant effect on the relative benefit of participating in P2P markets. Section V-B discusses the energy imbalances which households experience.

B. HOUSEHOLD ENERGY IMBALANCES

During individual settlement periods households can experience very high energy imbalances. The energy imbalance in a settlement period can be a significant proportion of, or even higher than the demand and generation. The largest energy imbalance in a settlement period, 6.713 kWh, is slightly less than the highest demand, 8.204 kWh, and much greater than the highest generation, 1.942 kWh.

However, these high energy imbalances are the extremities of a distribution with a very low mean. These large outlying values are reflected in the high kurtosis of the distributions, a measure of the size of the distribution's tails. The highest absolute mean energy imbalance of any household was only 0.0011 kWh.

Households have fairly equal energy imbalances in both the positive and negative direction. The energy imbalance distributions have a low skewness, a measure of the asymmetry of a distribution. Figure 2 shows the energy imbalance distribution of a typical household. This low skewness combined with the low mean causes the energy imbalances in the positive and negative direction to cancel out. Therefore, households have a low cumulative energy imbalance at the end of the run. The highest cumulative energy imbalance at the end of 10,000 settlement periods was 10.557 kWh, only 157% of the highest energy imbalance in a single settlement period.

Households which are making more accurate predictions of their energy supply and demand, experience lower energy imbalances. As the prediction accuracy of households improves (μ_{pred} decreases), the mean, highest and cumulative energy imbalances all decrease. This means that investing in technologies like energy storage which can reduce prediction errors will reduce energy imbalances of households. However, the shape of the energy imbalance distribution does not significantly change. The skewness and kurtosis of the energy imbalance distribution is not correlated with the prediction accuracy coefficient (μ_{pred}). Table 1 shows a summary of the energy imbalance distributions for different values of the prediction accuracy coefficient (μ_{pred}).

The closest mathematical approximation of the distribution observed in the results is Johnson's S_U distribution [57], [58]. Previous papers which have considered energy imbalances in



FIGURE 2. Probability density function of energy imbalance for a representative household acting under the current British imbalance charge mechanism ($\mu_{mal} = 1$, $\mu_{pred} = 1$).

P2P markets have assumed the imbalances follow a Gaussian distribution [38]. However, a Gaussian distribution is a relatively poor approximation of energy imbalances in the model presented in this paper. Table 2 in Appendix A gives the parameters for Johnson's S_U distribution which best represent the energy imbalances in the model for different values of prediction accuracy (μ_{pred}).

C. IMBALANCE CHARGES UNDER THE SYMMETRIC IMBALANCE CHARGE MECHANISM

Households subject to the symmetric imbalance charge mechanism incur positive imbalance charges which grow over time. Households are charged the imbalance price irrespective of the direction of their imbalance. Households subject to the symmetric imbalance mechanism at today's imbalance prices ($\mu_{imb} = 1$) and who do not take action to reduce their energy imbalance ($\mu_{pred} = 1$) have a distribution of imbalance charges with a large positive skew. The mean imbalance charge of these households was 0.08 GBP, with a minimum of -1.76 GBP and a maximum of 15.34 GBP. The reason it is possible for households to experience negative imbalance charges under the symmetric imbalance charge mechanism is that the imbalance price is occasionally negative.

The ratio of net to gross energy imbalance in the model is $\alpha = 0.4$ (see Equation 6). This means that reducing imbalance prices to 40% of today's level will keep the system operator's revenue constant. The value of $\alpha = 0.4$ is specific to the data used in this model. The system operator's revenue must be maintained to allow them to pay for the required balancing actions. Reducing the imbalance prices to 40% of today's level ($\mu_{imb} = 0.4$) reduces the imbalance charges households experience. The shape of the distribution does not change but the mean imbalance charge reduces to 0.03 GBP, and the maximum and minimum imbalance charges reduce to 16.80 GBP and -0.99 GBP respectively.

TABLE 1. Energy imbalance distribution summary data for all households.

		Highest		Highest				
		Absolute	Highest	Absolute	Range of			
Prediction		Mean	Absolute	Cumulative	Energy	Range of	Range of	
Accuracy		Energy	Energy	Energy	Imbalance	Energy	Energy	
Coeffic	cient	Imbalance	Imbalance	Imbalance	Standard	Imbalance	Imbalance	
(μ_{pre})	ed)	(kWh)	(kWh)	(kWh) Deviation		Skewness	Kurtosis	
	1.0	0.0011	7.244	10.557	0.034 - 1.282	-2.130 - 0.584	3.9 - 80.4	
	0.9	0.0007	5.603	6.702	0.050 - 1.031	-3.545 - 0.583	3.8 - 148.1	
JCe	0.8	0.0008	5.734	8.446	0.031 - 0.884	-1.994 - 0.923	3.8 - 89.7	
ng Ilar	0.7	0.0007	4.607	6.511	0.038 - 0.912	-2.248 - 0.820	3.8 - 87.3	
asi 1ba	0.6	0.0006	3.820	6.281	0.031 - 0.682	-1.671 - 1.201	3.8 - 113.8	
' in	0.5	0.0003	3.099	3.269	0.016 - 0.573	-1.994 - 0.674	4.1 - 122.9	
rg. De	0.4	0.0003	2.685	2.801	0.016 - 0.458	-1.994 - 0.781	3.8 - 47.6	
Iene	0.3	0.0002	1.974	2.512	0.015 - 0.395	-1.969 - 0.918	3.9 - 105.3	
9	0.2	0.0002	1.316	2.279	0.011 - 0.256	-2.130 - 0.986	3.8 - 80.4	
	↓ 0.1	0.0001	0.623	0.919	0.003 - 0.108	-1.916 - 0.583	4.3 - 113.8	



FIGURE 3. Cumulative imbalance charge over duration of the simulation for different values of imbalance price.

Under the symmetric imbalance charge mechanism, increasing the prediction accuracy (lowering energy imbalances) reduces the cumulative imbalance charges experiences by households over the duration of the simulation. Figure 3 shows how the cumulative imbalance charge develops over the duration of the simulation for the symmetric imbalance charge mechanism and an imbalance price multiplier $\mu_{pred} = 0.4$. The imbalance charge linearly increases over the duration of the model for the symmetric imbalance charge mechanism. As the prediction accuracy increases, the cumulative imbalance charge increases at a slower rate. This means that households have a financial incentive to invest in technologies such as storage which can reduce their energy imbalances.

D. HOUSEHOLD PROFITS UNDER THE SYMMETRIC IMBALANCE CHARGE MECHANISM

At today's imbalance price level ($\mu_{imb} = 1$), household energy imbalances must significantly decrease to allow P2P markets to function. When households are subject to the symmetric imbalance charge mechanism, current imbalance prices ($\mu_{imb} = 1$), and do not take action to reduce their energy imbalances ($\mu_{pred} = 1$), all households are worse off in the P2P market than the traditional market. This would cause households to switch out of the P2P market back to the traditional market. Reducing energy imbalances to 10% of their level when no action is taken ($\mu_{imb} = 0.1$) still means that 27.66% are worse off in the P2P market compared to the traditional market. Therefore, most households would have to significantly reduce their energy imbalance before being more profitable in a P2P market than a traditional market.

For a P2P market to function using the symmetric imbalance charge mechanism, the imbalance prices must be lower than they currently are in Great Britain. For this market, reducing imbalance prices to 40% of their current level ($\alpha = 0.4$) maintains the system operator's revenue at its current level. At $\mu_{imb} = 0.4$, 7.7% of households are better off in the P2P market when they do not take action to reduce their energy imbalances ($\mu_{pred} = 1$). If energy imbalances are reduced to 10% of their level when no action is taken ($\mu_{pred} = 0.1$), 90.2% of households are better off in the P2P market.

Figures 4 and 5 show box plots of the difference in profit between acting in a P2P and traditional market for imbalance price multipliers of $\mu_{imb} = 1$ and $\mu_{imb} = 0.4$ respectively. The plot for imbalance price multiplier of $\mu_{imb} = 1$ (Figure 4) shows that the prediction accuracy must be reduced to around 20% of its level when no action is taken before over 50% of households are better off in the P2P market. With an imbalance price multiplier of $\mu_{imb} = 0.4$, Figure 5 shows that a household only needs to reduce their prediction accuracy to 40% of the level when no action is taken before over 50% of households are better off in a P2P market. Table 3 in Appendix A shows the proportion of households which have a higher profit in the P2P market for different values of prediction accuracy and imbalance multiplier. Table 4 in Appendix A shows the mean additional profit when acting in the P2P market.



FIGURE 4. Box plot showing the difference in profit between the P2P market and the traditional market at the end of the simulation for imbalance price coefficient $\mu_{imb} = 1.0$.



FIGURE 5. Box plot showing the difference in profit between the P2P market and the traditional market at the end of the simulation for imbalance price coefficient $\mu_{imb} = 0.4$.

The imbalance price in Great Britain is currently the marginal cost of the balancing actions taken by the system operator. The reason the imbalance price is based on the marginal cost is that it incentivises BSC parties to continue to take action to balance their own supply and demand of energy until those actions become more costly than the same action taken by the system operator. This idea that the parties who can take balancing actions at the lowest cost should take those actions is based on minimising the overall system cost and therefore consumers bills. Reducing the imbalance price to 40% of its current level will leave a significant gap between the price of balancing actions which P2P market participants will take, and the cost of the system operator

taking those same actions. This will raise the overall system cost. Therefore, there is a trade off when setting the imbalance price between minimising imbalance charges for the P2P market participants and creating the correct incentives to reduce energy imbalances. Section V-H discusses the economic efficiency of the symmetric imbalance charge further.

Sections V-E and V-F will show that under the current British imbalance charge mechanism, P2P market participants are not incentivised to take any balancing actions. Since the imbalance prices in the positive and negative direction cancel out over time, imbalance charges are simply a cash flow issue for P2P market participants. The households may have to pay relatively large imbalance charges during a particular settlement period, however over time these large charges will be cancelled out by charges paid to the households by the system operator. Therefore, the current imbalance charge mechanism does not incentivise market participants to take balancing actions. Although the symmetric imbalance charge mechanism leaves a gap between the imbalance price and the system operator's marginal balancing action cost, it provides a greater incentive for P2P market participants to balance their supply and demand for energy than the current British imbalance charge mechanism does.

For any combination of μ_{imb} and μ_{pred} there is a range of results for household profit. This range affects whether or not households are better off in the P2P or traditional market. Some of this variability is explained by the ratio of installed demand capacity to installed generation capacity of the household, and by the market price. Households with a relatively high proportion of generation have higher profits at higher P2P market prices. Households with relatively high proportions of demand have higher profits at lower P2P market prices.

The proportion of generation to demand and the market price does not explain all the variance in profit. There is an element of randomness to the outcome of a household in the model because imbalance prices are not correlated to the energy imbalance in the model. Therefore two households with identical energy imbalances could end up with significantly different levels of profit under the symmetric imbalance charge mechanism. One household might happen to have high energy imbalances during settlement periods with low imbalance prices, leading to a lower imbalance charge. Another household might have high energy imbalance during settlement periods with high imbalances prices, leading to higher imbalance charges.

E. IMBALANCE CHARGES UNDER THE CURRENT BRITISH IMBALANCE CHARGE MECHANISM

The imbalance charges households experience when subject to the current British imbalance charge mechanism follow the same pattern as their energy imbalances. In individual settlement periods households can experience high imbalance charges. The highest imbalance charge during a settlement period for a household which did not take any action to reduce their energy imbalance ($\mu_{pred} = 1$) was 10.18 GBP. By comparison the highest revenue during a settlement period in the same group of households was 0.27 GBP. However, the distribution of imbalance charges has a low mean and skewness. The highest mean imbalance charge of any household under the current British imbalance charge mechanism was 0.006 GBP. Figure 6 shows the probability density function for the imbalance charges of a representative household acting under the current British imbalance charge mechanism.



FIGURE 6. Probability density function of imbalance charges for a representative household acting under the current British imbalance charge mechanism ($\mu_{mal} = 1$, $\mu_{pred} = 1$).

Over the course of the model run, the cumulative imbalance charge is low. Figure 7 shows the cumulative imbalance charge of the same representative household as Figure 6, developing over the course of the model run. These graphs show that the highest imbalance charge for this household during a single settlement period was 0.79 GBP. However, the cumulative imbalance charge at the end of 10,000 settlement period was -3.50 GBP (paid by the system operator to the household). The reason for this low cumulative imbalance charge is that the imbalance charges during individual settlement periods cancel out.

The reason the imbalance charges follow the same pattern as energy imbalances is that there is little correlation between energy imbalances and imbalance prices (Pearson coefficient $\rho = .01$). The reason for the poor correlation between energy imbalance and imbalance price is that there are currently no BSC parties who act like the households modelled in this paper. If the types of P2P markets modelled in this paper form a small part of the British electricity market, and therefore cause only a small proportion of the energy imbalances, this disconnect between energy imbalances in P2P markets and imbalance price is likely to remain. However, if P2P markets formed a substantial part of the electricity market, the system energy imbalance is likely to become correlated with the P2P market energy imbalance.



FIGURE 7. Graph of cumulative imbalance charges for a representative household acting under the current British imbalance charge mechanism ($\mu_{mal} = 1$, $\mu_{pred} = 1$).

A higher correlation between energy imbalance and imbalance price could lead to higher cumulative imbalance charges. It would become more likely that the energy imbalance of the household would be in the same direction as the system imbalance. When a household's energy imbalance is in the same direction as the system imbalance they are penalised. Therefore, the mean energy imbalance of a household would increase, leading to a higher cumulative imbalance charge over time. The extent to which higher correlations between energy imbalance and imbalance price would affect the imbalance charge is not considered in this paper.

F. HOUSEHOLD PROFIT UNDER THE CURRENT BRITISH IMBALANCE CHARGE MECHANISM

Under the current British imbalance charge mechanism, 98.6% of households (491 of 498) which do not act maliciously have a higher profit in the P2P market than the traditional market. This level of profitability is caused by the fact that households have higher revenues in the P2P market and experience low imbalance charges, see Sections V-A and V-E respectively. Although this level of profitability in P2P markets is beneficial to the households, it is indicative of the fact that the current British imbalance charge mechanism does not provide an incentive for households to reduce their energy imbalances.

The level of energy imbalance does not affect the profitability of households. Changing the prediction accuracy, μ_{pred} , has little effect on the profitability of households. For high values of μ_{pred} households experience higher imbalance charges in any given settlement period. At $\mu_{pred} = 1$ the average absolute imbalance charge is 0.08 GBP, compared to 0.01 GBP when $\mu_{pred} = 0.1$. However, the imbalance charges are equal in opposite directions. Over enough settlement periods the imbalance charges average out to low values irrespective of the value of μ_{pred} . For example at $\mu_{pred} = 1$, the average total imbalance charge at the end of the model run was -14.33 GBP, compared to -2.19 GBP for $\mu_{pred} = 0.1$.

Figure 8 shows a box plot of the difference in profit between the P2P market and the traditional market at the end of the simulation. Almost all households have a higher average profit in the P2P market. The prediction accuracy coefficient does not affect the household profit.



FIGURE 8. Box plot showing the difference in profit between the peer-to-peer market and the traditional market at the end of the simulation.

Not addressing the high potential energy imbalances in P2P electricity markets could create a problem for the whole electricity network, and for P2P markets internally. The system operator must remedy any energy imbalances to ensure there is sufficient energy to meet the network demand. The imbalance price is calculated using the marginal cost of balancing actions during each settlement period. Therefore, higher total system energy imbalances will lead to higher imbalance prices, which in turn will be passed onto the consumers, both inside and outside the P2P market, as higher energy prices. The mean ratio of energy imbalance to demand in the model is 1.007. In the current British electricity network, the mean energy imbalance to demand ratio in 2020 was 0.010 [43]. It is not possible to directly compare these two figures because the imbalances at a national level will net-off. However, they do show that imbalances in P2P market are likely to be very high relative to the market demand.

High imbalances within P2P markets might cause a problem for the internal functioning of those markets. Many P2P market designs consist of an optimisation problem in a futures market. Deviation from the futures market position, an energy imbalance, will reduce the efficiency of these markets. Sections V-C and V-D present the results of the symmetric imbalance charge mechanism and shows how it can alleviate this problem.

G. MALICIOUS BEHAVIOUR

A secondary advantage of the symmetric imbalance charge mechanism is it prevents households from manipulating their final notification volume, E_{fnv} , to increase their profit. Two types of malicious behaviour have been modelled. They are described in Section III-B. The results presented in this section demonstrate that under the current British imbalance charge mechanism, households can manipulate their final notification volume to increase their own profit to the detriment of the electricity system. The results also show that under the symmetric imbalance charge mechanism households cannot increase their profit by manipulating their final notification volume.

1) PURCHASING MORE ENERGY THAN REQUIRED

In the model, 2,004 of the 10,000 households acted maliciously by decreasing their final notification volume, E_{fiv} , by 50%. Of the households acting maliciously in that manner, 165 were subjected to the current British imbalance charge mechanism. The mean profit of these 165 household was 189.58 GBP/year. By contrast, the mean profit was -44.59 GBP/year for households which did not manipulate their final notification volume. This additional profit gives households a significant incentive to act dishonestly in the market.

The reason the household's profit is increased by reducing the final notification volume is that it increases the likelihood the household will export more energy than the final notification volume. This increases the likelihood the household will be paid the imbalance charge. The mean imbalance charge for households not acting maliciously was -12.98 GBP/year. However, for households acting maliciously the mean imbalance charge was -518.61 GBP/year, where a negative charge is paid to the household by the system operator. This significant increase in the imbalance charge paid to the household dramatically increases their profit.

Acting maliciously actually reduces the revenue of the household in the P2P market. The mean revenue of households not acting maliciously was -57.57 GBP/year, compared to -329.02 GBP/year for households acting maliciously. However, the reduction in revenue was more than offset by the additional imbalance charge paid to the household, leading to a higher profit.

The reason it is undesirable for households to adjust their final notification volume is that it dramatically increases the energy imbalance in the market. The mean energy imbalance of households not acting maliciously was 0.00002 kWh. By contrast, households acting maliciously by decreasing their final notification volume had a mean energy imbalance of 0.07 kWh. The high energy imbalances of one market participant are detrimental to all other energy market participants, as discussed in Section V-C. The symmetric imbalance charge mechanism removes this incentive for households to artificially increase their energy imbalance.

The symmetric imbalance charge eliminates the benefit of manipulating the final notification volume. Under the symmetric imbalance charge mechanism, the mean profit of households ($\mu_{imb} = 1$) acting maliciously was -1,220.40 GBP/year, compared to -766.41 GBP/year for households acting honestly. Acting maliciously leaves households considerably worse off than acting honestly. Acting maliciously does not have the same impact on the imbalance charges households incur under the symmetric imbalance charge mechanism. The mean imbalance charge for households acting honestly under the symmetric imbalance charge regime ($\mu_{imb} = 1$) was 682.42 GBP/year, compared to 911.11 GBP/year for households acting maliciously. Therefore, the symmetric imbalance charge mechanism incentivises households to accurately report their final notification volumes, decreasing the total system energy imbalance.

This section has shown that the symmetric imbalance charge mechanism eliminates the possibility of households increasing their profit by decreasing their final notification volume. The next section will show that the symmetric imbalance charge mechanism also removes the benefit of the more sophisticated form of malicious behaviour based on predicting imbalance prices.

2) ADJUSTING POSITION BASED ON IMBALANCE PRICE PREDICTION

If a household is able to predict whether the market price or imbalance price will be higher, they can increase their profits under the current British imbalance charge mechanism by adjusting their final notification volume, E_{fnv} . A full description of the method of malicious behaviour is presented in Section III-B2. In the model 2,068 of 10,000 households acted maliciously in this manner. Of those 2,068 households, 167 were subject to the current British imbalance charge mechanism. The mean profit of those 167 households was 244.17 GBP/year. This profit is higher than the mean profit of households acting in the less sophisticated malicious manner described in Section V-G1 (189.58 GBP/year), and significantly higher than households not acting maliciously (-44.59 GBP/year).

The imbalance charge of households acting maliciously by predicting the market and imbalance prices is much lower (more money paid to the household) than those which are not under the current British imbalance charge mechanism. The mean imbalance charge was -12.98 GBP/year for households not acting maliciously, compared to -507.60 GBP/year for those predicting the market and imbalance prices. Revenues are lower for households which adjust their final notification volumes, -263.43 GBP/year as opposed to -57.57 GBP/year for those which do not, but that is not sufficient to offset the gains made from imbalance charges.

As with the previous method of malicious action, the symmetric imbalance charge mechanism eliminates the benefit of acting maliciously in this manner. The profit of households acting maliciously by predicting the market and imbalance prices under the symmetric imbalance charge ($\mu_{imb} = 1$) was

-1106.99 GBP/year, compared to -766.41 GBP/year for those not acting maliciously. This significantly lower profit will disincentivise households from acting maliciously. This decrease in profit for households acting maliciously under the symmetric imbalance charge is caused by a higher imbalance charge (836.01 GBP/year for households acting maliciously compared to 682.42 GBP/year for those not), combined with a lower revenue (-270.98 GBP/year for households acting maliciously compared to -84.00 GBP/year for those not).

The results presented in this section have shown that the symmetric imbalance charge produces a financial penalty if households choose to adjust their final notification volume from their best prediction of energy supply and demand, artificially creating an energy imbalance. This contrasts with the current British imbalance charge mechanism, under which households can financially gain by adjusting their final notification volume, giving them an incentive to artificially increase their energy imbalances.

H. ECONOMIC EFFICIENCY OF THE SYMMETRIC IMBALANCE CHARGE MECHANISM

The current British imbalance price is set at the marginal cost of the balancing mechanism during each settlement period. Setting the imbalance price at the marginal cost of the balancing mechanism should incentivise market participants who can take balancing actions more cheaply than the system operator. Under the symmetric imbalance charge we have suggested it is possible, but not necessary for the imbalance price to be lowered. See Sections II and III for more details on the imbalance price in Britain now and under the symmetric imbalance charge mechanism.

Reducing the imbalance price below the marginal cost of the balancing mechanism creates the possibility of economic inefficiency. The system operator might be forced to take a balancing action which a market participant could have taken more cheaply. However, because the price of that balancing action was above the imbalance price, the market participant would have been worse off by taking it. However, this economic efficiency makes two assumptions which may not be correct in P2P markets.

Firstly, the economic efficiency argument assumes that market participants know what the imbalance price will be. In Britain the imbalance price is only published after the settlement period [43]. For market participants to act on the imbalance price they must make a prediction of what it is likely to be. Households are less likely to be able to make such predictions than sophisticated traders in traditional markets.

The second important assumption made by the economic efficiency argument is that market participants will be able to respond to the imbalance price. The main thing which determines the ability of P2P market participants to respond to the imbalance price is the level of storage and flexible load in the market. Storage and flexible loads have low operating costs but high capital costs. The imbalance charge in a single settlement period, which will determine how storage and flexible loads are operated, is therefore less important than the cumulative imbalance charge over time, which will determine the investment in storage and flexible loads.

The results presented in Section V-E show that the cumulative imbalance charge of a household subjected to the current British imbalance charge mechanism will be very low. This low cumulative imbalance charge provides no incentive for them to invest in storage or flexible loads. Therefore, they will be unable to respond to high imbalance prices during any particular settlement period.

In contrast, the results presented in Section V-C show that the cumulative imbalance charge of a household subjected to the symmetric imbalance charge mechanism consistently grows over time. This cumulative imbalance charge growth provides a strong incentive for investment in storage and flexible loads. It is therefore likely that although households have a less economically efficient incentive to adjust their energy imbalances during a settlement period, they will have a more effective incentive to reduce their energy imbalances over time.

I. IMPLEMENTATION OF THE SYMMETRIC IMBALANCE CHARGE MECHANISM

There are two ways the symmetric imbalance charge mechanism could be implemented. Firstly, the codes which govern imbalance charges (the BSC in Great Britain) could be amended so that all market participants, both inside and outside the P2P market, are subject to the symmetric imbalance charge mechanism. Secondly, a P2P market operator could impose the symmetric imbalance charge mechanism only within a P2P market.

Subjecting the whole electricity market to the symmetric imbalance charge mechanism could have significant consequences for traditional market participants. The business models and trading arrangements of current physical and non-physical traders are designed around the current system. A significant change, as proposed in this paper, would have ramifications for them which would need to be assessed. Further evidence of the impact of these changes on current BSC parties would need to be gathered before a change to the whole system could be considered. In Great Britain, Ofgem has a significant code review process which allows them to assess the effect of changes on BSC parties.

Only subjecting a P2P market to the symmetric imbalance charge mechanism would require minimal to no changes to codes, and would not have any impact on traditional market participants. A P2P market operator could act as the conduit to the traditional market (BSC party in Great Britain) for the whole P2P market. The market operator would be subject to the imbalance charges for all P2P market participants under the current imbalance charge mechanism. The market operator can then impose the symmetric imbalance charge mechanism on the P2P market participants. This paper has shown how the revenue of the two imbalance price mechanisms can be made equal. Therefore, the market operator will receive the same revenue from the P2P market participant as they must pay in imbalance changes to the system operator. In this way the P2P market participants are being effectively subjected to the symmetric imbalance charge mechanism with minimal effect to the rest of the electricity market. In Great Britain the P2P market operator may be able to act as a BSC party under the Virtual Power Plant provision in the current BSC [3].

In the event the symmetric imbalance charge is only implemented inside a P2P market, the market operator must choose to impose it upon the P2P market. The most likely reason the system operator would choose to implement an imbalance charge system that incentivises reduced imbalances is that imbalances make the P2P market less efficient. Many P2P market designs are based around an optimisation function in a futures market. Energy imbalances are not accounted for in the optimisations and therefore make the P2P market less efficient.

J. METHODS OF REDUCING ENERGY IMBALANCES IN PEER-TO-PEER MARKETS

The results presented in Section V-D show that the symmetric imbalance charge mechanism gives P2P market participants a clear financial incentive to reduce their energy imbalances. This section briefly discussed the methods which P2P market participants could employ to reduce their energy imbalances. These methods fall into three broad categories: improving energy supply and demand predictions; increasing flexibility of energy supply and demand; and netting-off energy imbalances within a P2P market.

Energy imbalances of P2P market participants who are not acting maliciously are caused by errors in energy supply and demand forecasting. Improving the accuracy of the forecasts used, e.g. through improved weather forecasting, will reduce the level of energy imbalance.

Installing sources of flexibility, such as energy storage or flexible loads, will allow P2P market participants to adjust their supply or demand for energy during a settlement period. These supply and demand adjustments can be used to reduce energy imbalances. In the circumstance where a whole P2P market is a single BSC party, real-time trading can be used to take advantage of all the storage and flexible load in the market. If one market participant is experiencing an energy imbalance during a settlement period, they can sell or purchase energy from other participants using storage or flexible load to adjust their own supply or demand. To the system operator this will just appear to be a reduction in energy imbalance.

If the whole P2P market is a single BSC party and two market participants have energy imbalances in opposite directions, those participants can trade to eliminate both their energy imbalances. This possibility of P2P trading as a means of reducing imbalances has been discussed in other works [28], [29]. This is possible because all imbalances within the market are netted-off from the system operator's point of view as the whole market acts as a single BSC party.

VI. CONCLUSION

This paper has proposed a symmetric imbalance charge mechanism which provides a financial incentive for P2P electricity market participants to reduce their energy imbalances. Having a financial incentive is important because the measures a market participant might take to reduce their energy imbalances are expensive. These measures include installing energy storage or demand response, or trading with other market participants who have storage or demand response. The symmetric imbalance charge would therefore complement P2P market designs which use such devices with the aim of minimising energy imbalances, by providing an external financial incentive. The symmetric imbalance charge mechanism also prevents P2P market participants from adjusting their final notification volumes to benefit from the imbalance charge mechanism at the expense of other electricity market participants.

To show the advantages of the symmetric imbalance charge mechanism, this paper has shown that the current British imbalance charge mechanism does not provide an effective incentive for P2P market participants to reduce their energy imbalances. The energy imbalances of households who might participate in P2P markets are much higher, as a proportion of demand, than the energy imbalances currently experienced in Great Britain. Therefore, the introduction of P2P electricity markets without an incentive for participants to reduce their energy imbalances might significantly increase the total energy imbalance, and therefore overall electricity system cost.

Finally, this paper has addressed some of the issues involved with implementing the symmetric imbalance charge mechanism. The imbalance price can be reduced while maintaining the system operator's revenue. The symmetric imbalance charge mechanism could either be implemented as a change to the British BSC, or it could be implemented only within the boundaries of a P2P electricity market.

The work in this paper has shown that when P2P market participants are subject to appropriate imbalance charges, the charges are not negligible. Therefore, P2P market trading strategies should consider imbalance charges. Most current literature proposing P2P trading strategies neglects imbalance charges. Future work should consider what effect these imbalance charges have on optimal trading strategies in P2P markets, and on P2P market design.

This work has also shown that P2P markets could have a significant effect on the energy imbalance of the electricity network. Therefore, the system operator and regulators should ensure that codes and standards effectively incentivise P2P markets to minimise their imbalances. The introduction of P2P markets is likely to require an amendment to the British BSC and other similar codes in different countries. Future work should consider the design of this amendment, and the effect of this amendment on traditional electricity market participants such as energy suppliers. In this paper we have made several key assumptions, including that the household can buy and sell all the energy they require in the P2P market, and that the P2P market energy imbalances do not affect the imbalance price. These assumptions limit how broadly the results of this paper can be applied. Future work could usefully examine the effect of these assumptions by applying imbalance charges to P2P market models with a finite supply and demand for energy, and by examining the size of P2P markets that would begin to affect imbalance charges.

The work presented in this paper demonstrates that, for peer-to-peer markets to function, the design of imbalance charge must be carefully considered. The regulatory challenges associated with energy imbalances in P2P markets are surmountable. They should not stand in the way of the environmental benefits large scale adoption of distributed renewable generators can bring. We have demonstrated a simple alternative imbalance charge design which provides peer-to-peer market participants with corrective incentives to reduce their energy imbalances.

DATA AVAILABILITY

This study is an analysis of three existing publicly available datasets which are openly available from the Elexon Portal at https://www.elexonportal.co.uk, and the London Datastore at https://data.london.gov.uk/dataset/smartmeter-energy-use-data-in-london-households and https://data.london.gov.uk/ dataset/photovoltaic_pv-solar-panel-energy-generation-data, reference numbers [43], [45], [46]. The code for the model developed in this study is openly available in Figshare at https://doi.org/10.48420/14681502, reference number [59].

APPENDIX A

SUPPORTING DATA

This appendix contains supporting numerical results. In addition to the supporting data in this section, the location of the datasets and code used in this paper can be found in Section VI.

TABLE 2. Johnson's S	u distribution	parameters for	or energy	imbalances.
----------------------	----------------	----------------	-----------	-------------

Predic	tion									
Accuracy										
C ff			, a b'							
Coeffic	eient	Johnson's S_U Distribution Parameters								
$(\mu_{pre}$	(d)	γ	δ	ξ	λ					
	1.0	0.01036806	0.5625183	0.004420315	0.06183279					
	0.9	0.009466865	0.547397	0.003673008	0.0499467					
JCe	0.8	0.008295291	0.5533258	0.003397922	0.04700802					
ng Ilar	0.7	0.008781784	0.5586379	0.002746877	0.04178138					
asi	0.6	0.01073683	0.5601633	0.002548438	0.03611655					
' in	0.5	0.009438837	0.5476537	0.001941413	0.02788683					
De rgy	0.4	0.009319987	0.5656411	0.001584124	0.02520433					
l	0.3	0.008683168	0.5609285	0.001211573	0.0184287					
9	0.2	0.01094719	0.554656	0.0008810541	0.01131903					
	↓ 0.1	0.009949628	0.5509336	0.0004090998	0.005739197					

Table 2 shows the Johnson's S_U distribution parameters for different values of the prediction accuracy coefficient (μ_{pred}). These parameters have been calculated to give the best fit for

TABLE 3. Decimal proportion of households with higher profits in the P2P market.

			Prediction Accuracy Coefficient (μ_{pred})										
				Increasing energy imbalance									
	0.0 0.1 0.2 0.3 0.4 0.5 0.6 0.7 0.8 0.9												1.0
<i>(q</i>	icreasing prices	0.0	1.00	1.00	1.00	1.00	0.96	1.00	1.00	1.00	1.00	1.00	0.98
efficient (μ_{im}		0.1	1.00	0.93	0.89	0.87	0.89	0.84	0.81	0.77	0.74	0.74	0.72
		0.2	1.00	0.83	0.89	0.83	0.79	0.71	0.59	0.52	0.47	0.50	0.41
		0.3	1.00	0.81	0.77	0.79	0.65	0.59	0.44	0.37	0.31	0.23	0.13
		0.4	1.00	0.95	0.77	0.58	0.51	0.36	0.17	0.26	0.14	0.18	0.08
		0.5	1.00	0.87	0.60	0.51	0.30	0.26	0.20	0.13	0.03	0.08	0.02
ပိ		0.6	1.00	0.82	0.71	0.39	0.22	0.17	0.07	0.00	0.03	0.00	0.00
balance		0.7	1.00	0.85	0.53	0.24	0.21	0.09	0.10	0.07	0.00	0.02	0.00
	Ir	0.8	0.98	0.78	0.55	0.16	0.14	0.10	0.02	0.02	0.00	0.00	0.00
		0.9	1.00	0.81	0.44	0.15	0.14	0.05	0.00	0.00	0.00	0.00	0.00
Im	`	1.0	1.00	0.57	0.38	0.12	0.07	0.03	0.02	0.00	0.00	0.00	0.00

TABLE 4. Additional profit of households in the P2P market (GBP per year, positive is additional profit in P2P market).

		Prediction Accuracy Coefficient (μ_{pred})											
			Increasing energy imbalance									\ \	
			0.0	0.1	0.2	0.3	0.4	0.5	0.6	0.7	0.8	0.9	1.0
$\operatorname{ent}(\mu_{imb})$		0.0	291.99	317.32	180.75	293.57	269.56	192.78	238.69	253.22	255.34	180.23	215.68
		0.1	251.99	205.08	256.86	233.70	193.74	222.45	164.23	173.37	126.68	97.54	89.45
	s	0.2	265.72	153.45	192.94	167.14	125.57	51.00	53.03	52.84	-12.56	21.53	-19.53
	ice	0.3	261.70	168.98	118.25	174.78	79.91	85.96	-18.99	-63.06	-96.89	-127.13	-149.20
ici,	easing pr	0.4	230.55	176.87	119.89	48.73	14.88	-42.13	-141.44	-120.64	-228.71	-220.65	-359.79
eff		0.5	207.97	156.14	66.28	0.60	-74.38	-94.82	-195.08	-240.52	-417.63	-475.41	-537.14
balance Co		0.6	222.47	149.24	81.56	-36.61	-119.44	-200.36	-280.89	-336.75	-465.42	-399.94	-658.66
	lcre	0.7	179.48	150.67	25.21	-58.02	-96.33	-268.21	-325.96	-385.24	-530.33	-596.20	-705.30
	Ц	0.8	268.15	105.94	36.51	-138.44	-191.56	-330.71	-411.16	-630.76	-500.03	-739.21	-944.35
		0.9	238.09	136.41	-8.32	-148.76	-269.70	-335.25	-570.25	-617.40	-729.28	-729.31	-1060.22
Im		1.0	222.13	42.65	-82.79	-183.44	-319.02	-372.06	-550.64	-752.19	-903.39	-1079.69	-1109.78

the total distribution of energy imbalances for all households with each value of μ_{pred} who are not acting maliciously.

Table 3 shows the decimal proportion of households who have a higher profit in a P2P market compared to a traditional market for different values of prediction accuracy coefficient (μ_{pred}) and imbalance charge coefficient (μ_{imb}). The prediction accuracy coefficient (μ_{pred}) is multiplied by the energy imbalance, meaning low values of μ_{pred} are associated with lower energy imbalances and higher values with higher energy imbalances. Likewise, the imbalance charge coefficient (μ_{imb}) is multiplied by the imbalance price, so low values of μ_{imb} are associated with lower imbalance prices and higher values with higher imbalance prices.

Table 4 shows the average difference in profit between the P2P and traditional markets (values in pence). A positive value means the profit was higher in the P2P market, a negative value means the profit was higher in the traditional market.

ACKNOWLEDGMENT

The authors would like to thank Prof. Sydney Howell of the Alliance Manchester Business School, University of Manchester, for his very helpful comments on the design of this work.

REFERENCES

 Department for Business Energy & Industrial Strategy. (2021). Solar Photovoltaics Deployment. [Online]. Available: https://www.gov.uk/ government/statistics/solar-photovoltaics-deployment

- [2] Ofgem. (2019). Smart Export Guarantee: Guidance for Generators. [Online]. Available: https://www.ofgem.gov.uk/system/files/docs/ 2020/02/seg_generator_guidance_-_final_for_publication.pdf
- [3] Elexon. (2020). Balancing & Settlement Code. [Online]. Available: https://www.elexon.co.uk/bsc-and-codes/balancing-settlement-code/ https://www.elexon.co.uk/bsc-and-codes/balancing-settlement-code/
- [4] Elia. *The Role of the BRP*. Accessed: Feb. 14, 2022. [Online]. Available: https://www.elia.be/en/electricity-market-and-system/role-of-brp
- [5] TenneT. Balance Responsible Parties. Accessed: Feb. 14, 2022. [Online]. Available: https://www.tennet.eu/our-key-tasks/energy-industry/balanceresponsible-parties/
- [6] PJM. (2020). PJM Manual 12: Balancing Operations. [Online]. Available: https://www.pjm.com/-/media/training/nerc-certifications/trans-exammaterials-2020/manuals-user-guides/balancing-operations.ashx
- [7] D. Brown, S. Hall, and M. E. Davis, "Prosumers in the post subsidy era: An exploration of new prosumer business models in the UK," *Energy Policy*, vol. 135, Dec. 2019, Art. no. 110984.
- [8] T. Capper, A. Gorbatcheva, M. A. Mustafa, M. Bahloul, J. M. Schwidtal, R. Chitchyan, M. Andoni, V. Robu, M. Montakhabi, I. J. Scott, C. Francis, T. Mbavarira, J. M. Espana, and L. Kiesling, "Peer-to-peer, community self-consumption, and transactive energy: A systematic literature review of local energy market models," *Renew. Sustain. Energy Rev.*, vol. 162, Jul. 2022, Art. no. 112403.
- [9] L. de Almeida, V. Cappelli, N. Klausmann, and H. van Soest, "Peer-to-peer trading and energy community in the electricity market—Analysing the literature on law and regulation and looking ahead to future challenges," Eur. Univ. Inst., Fiesole, Italy, Tech. Rep. RSC 2021/35, 2021. [Online]. Available: https://cadmus. eui.eu/bitstream/handle/1814/70457/RSC%202021_35rev.pdf?sequence= 4&isAllowed=y
- [10] W. Tushar, T. K. Saha, C. Yuen, M. I. Azim, T. Morstyn, H. V. Poor, D. Niyato, and R. Bean, "A coalition formation game framework for peer-to-peer energy trading," *Appl. Energy*, vol. 261, Mar. 2020, Art. no. 114436.
- [11] M. S. H. Nizami, M. J. Hossain, and E. Fernandez, "Multiagent-based transactive energy management systems for residential buildings with distributed energy resources," *IEEE Trans. Ind. Informat.*, vol. 16, no. 3, pp. 1836–1847, Mar. 2020.

- [12] Y. Wang, X. Wu, Y. Li, R. Yan, Y. Tan, X. Qiao, and Y. Cao, "Autonomous energy community based on energy contract," *IET Gener., Transmiss. Distribut.*, vol. 14, no. 4, pp. 682–689, Feb. 2020.
- [13] D. L. Rodrigues, X. Ye, X. Xia, and B. Zhu, "Battery energy storage sizing optimisation for different ownership structures in a peer-to-peer energy sharing community," *Appl. Energy*, vol. 262, Mar. 2020, Art. no. 114498.
- [14] M. Khorasany, Y. Mishra, and G. Ledwich, "A decentralized bilateral energy trading system for peer-to-peer electricity markets," *IEEE Trans. Ind. Electron.*, vol. 67, no. 6, pp. 4646–4657, Jun. 2020.
- [15] W. Amin, Q. Huang, M. Afzal, A. A. Khan, K. Umer, and S. A. Ahmed, "A converging non-cooperative & cooperative game theory approach for stabilizing peer-to-peer electricity trading," *Electr. Power Syst. Res.*, vol. 183, Jun. 2020, Art. no. 106278.
- [16] C. Feng, Z. Li, M. Shahidehpour, F. Wen, and Q. Li, "Stackelberg game based transactive pricing for optimal demand response in power distribution systems," *Int. J. Electr. Power Energy Syst.*, vol. 118, Jun. 2020, Art. no. 105764.
- [17] J. P. Palacios, M. E. Samper, and A. Vargas, "Dynamic transactive energy scheme for smart distribution networks in a Latin American context," *IET Gener., Transmiss. Distrib.*, vol. 13, no. 9, pp. 1481–1490, May 2019.
- [18] X. Wen, D. Abbes, and B. Francois, "Modeling of photovoltaic power uncertainties for impact analysis on generation scheduling and cost of an urban micro grid," *Math. Comput. Simul.*, vol. 183, pp. 116–128, May 2021.
- [19] B. Zhou, Y. Meng, W. Huang, H. Wang, L. Deng, S. Huang, and J. Wei, "Multi-energy net load forecasting for integrated local energy systems with heterogeneous prosumers," *Int. J. Electr. Power Energy Syst.*, vol. 126, Mar. 2021, Art. no. 106542.
- [20] Ofgem. (2021). Regulatory Sandbox Repository. [Online]. Available: https://www.ofgem.gov.uk/publications/regulatory-sandbox-repository
- [21] (2020). Innovation Sandbox Service—Overview. [Online]. Available: https://www.ofgem.gov.uk/sites/default/files/docs/2020/02/sandbox _service_overview.pdf
- [22] Z. Zhang, R. Li, and F. Li, "A novel peer-to-peer local electricity market for joint trading of energy and uncertainty," *IEEE Trans. Smart Grid*, vol. 11, no. 2, pp. 1205–1215, Mar. 2020.
- [23] M. H. Shams, M. Shahabi, and M. E. Khodayar, "Stochastic day-ahead scheduling of multiple energy carrier microgrids with demand response," *Energy*, vol. 155, pp. 326–338, Jul. 2018.
- [24] A. Baringo, L. Baringo, and J. M. Arroyo, "Day-ahead self-scheduling of a virtual power plant in energy and reserve electricity markets under uncertainty," *IEEE Trans. Power Syst.*, vol. 34, no. 3, pp. 1881–1894, May 2019.
- [25] D. Q. Hung, N. Mithulananthan, and K. Y. Lee, "Determining PV penetration for distribution systems with time-varying load models," *IEEE Trans. Power Syst.*, vol. 29, no. 6, pp. 3048–3057, Nov. 2014.
- [26] K. Anoh, S. Maharjan, A. Ikpehai, Y. Zhang, and B. Adebisi, "Energy peer-to-peer trading in virtual microgrids in smart grids: A game-theoretic approach," *IEEE Trans. Smart Grid*, vol. 11, no. 2, pp. 1264–1275, Mar. 2020.
- [27] M. S. H. Nizami, M. J. Hossain, B. M. R. Amin, and E. Fernandez, "A residential energy management system with bi-level optimizationbased bidding strategy for day-ahead bi-directional electricity trading," *Appl. Energy*, vol. 261, Mar. 2020, Art. no. 114322.
- [28] Y. Chen, S. Mei, F. Zhou, S. H. Low, W. Wei, and F. Liu, "An energy sharing game with generalized demand bidding: Model and properties," *IEEE Trans. Smart Grid*, vol. 11, no. 3, pp. 2055–2066, May 2020.
- [29] S. Chakraborty, T. Baarslag, and M. Kaisers, "Automated peer-to-peer negotiation for energy contract settlements in residential cooperatives," *Appl. Energy*, vol. 259, Feb. 2020, Art. no. 114173.
- [30] W. Tushar, T. K. Saha, C. Yuen, T. Morstyn, N.-A. Masood, H. V. Poor, and R. Bean, "Grid influenced peer-to-peer energy trading," *IEEE Trans. Smart Grid*, vol. 11, no. 2, pp. 1407–1418, Mar. 2020.
- [31] M. L. Di Silvestre, P. Gallo, M. G. Ippolito, R. Musca, E. R. Sanseverino, Q. T. T. Tran, and G. Zizzo, "Ancillary services in the energy blockchain for microgrids," *IEEE Trans. Ind. Appl.*, vol. 55, no. 6, pp. 7310–7319, Nov. 2019.
- [32] N. K. Meena, J. Yang, and E. Zacharis, "Optimisation framework for the design and operation of open-market urban and remote community microgrids," *Appl. Energy*, vol. 252, Oct. 2019, Art. no. 113399.
- [33] A. Lüth, J. M. Zepter, P. C. D. Granado, and R. Egging, "Local electricity market designs for peer-to-peer trading: The role of battery flexibility," *Appl. Energy*, vol. 229, pp. 1233–1243, Nov. 2018.

- [34] J. Qiu, K. Meng, Y. Zheng, and Z. Y. Dong, "Optimal scheduling of distributed energy resources as a virtual power plant in a transactive energy framework," *IET Gener., Transmiss. Distrib.*, vol. 11, no. 13, pp. 3417–3427, Sep. 2017.
- [35] Élexon. (2020). *Imbalance Pricing Guidance*. [Online]. Available: https://www.elexon.co.uk/documents/training-guidance/bsc-guidancenotes/imbalance-pricing/
- [36] Ofgem. (2020). Finding That Intergen has Breached Article 5 (Prohibition on Market Manipulation) of Regulation (EU) No 1227/2011 of the European Parliament and of the Council of 25 October 2011 on Wholesale Energy Market Integrity and Transparency ('Remit'). [Online]. Available: https://www.ofgem.gov.uk/publications/finding-intergen-hasbreached-article-5-prohibition-market-manipulation-regulation-eu-no-12272011-european-parliament-and-council-25-october-2011-wholesaleenergy-market-integrity-and-transparency-remit
- [37] Elexon. (2020). BSC Signatories and Qualified Persons. [Online]. Available: https://www.elexon.co.uk/bsc-and-codes/bsc-signatories-qualifiedpersons/
- [38] R. Ghorani, M. Fotuhi-Firuzabad, and M. Moeini-Aghtaie, "Main challenges of implementing penalty mechanisms in transactive electricity markets," *IEEE Trans. Power Syst.*, vol. 34, no. 5, pp. 3954–3956, Sep. 2019.
- [39] C. Möller, S. T. Rachev, and F. J. Fabozzi, "Balancing energy strategies in electricity portfolio management," *Energy Econ.*, vol. 33, no. 1, pp. 2–11, Jan. 2011.
- [40] R. A. C. van der Veen, A. Abbasy, and R. A. Hakvoort, "Agent-based analysis of the impact of the imbalance pricing mechanism on market behavior in electricity balancing markets," *Energy Econ.*, vol. 34, no. 4, pp. 874–881, Jul. 2012.
- [41] J. P. Chaves-Ávila, R. A. Hakvoort, and A. Ramos, "The impact of European balancing rules on wind power economics and on short-term bidding strategies," *Energy Policy*, vol. 68, pp. 383–393, May 2014.
- [42] S. Just and C. Weber, "Strategic behavior in the German balancing energy mechanism: Incentives, evidence, costs and solutions," J. Regulatory Econ., vol. 48, no. 2, pp. 218–243, Oct. 2015.
- [43] Elexon. (2020). *Elexon Portal*. [Online]. Available: https://www.elexonportal.co.uk
- [44] S. Noor, W. Yang, M. Guo, K. H. van Dam, and X. Wang, "Energy demand side management within micro-grid networks enhanced by blockchain," *Appl. Energy*, vol. 228, pp. 1385–1398, Aug. 2018, doi: 10.1016/j.apenergy.2018.07.012.
- [45] UK Power Networks. (2015). Smart Meter Energy Consumption Data in London Households. https://data.london.gov.uk/dataset/smartmeterenergy-use-data-in-london-households
- [46] (2017). Photovoltaic (PV) Solar Panel Energy Generation Data. [Online]. Available: https://data.london.gov.uk/dataset/photovoltaic-pvsolar-panel-energy-generation-data
- [47] European Commission. (2019). Photovoltaic Geographical Information System. [Online]. Available: https://re.jrc.ec.europa.eu/pvg_tools/ en/tools.html
- [48] Ofgem. (2014). Electricity Balancing Significant Code Review—Final Policy Decision. https://www.ofgem.gov.uk/sites/default/files/docs/2014/05/ electricity_balancing_significant_code_review_-_final_policy_decision.pdf
- [49] UK Power. (2020). Compare Energy Prices Per kWh. [Online]. Available: https://www.ukpower.co.uk/home_energy/tariffs-per-unit-kwh
- [50] Solar Guide. (2020). Compare Smart Export Guarantee Tariffs. [Online]. Available: https://www.solarguide.co.uk/smart-export-guarantee -comparison#/
- [51] N. Metropolis, A. W. Rosenbluth, M. N. Rosenbluth, A. H. Teller, and E. Teller, "Equation of state calculations by fast computing machines," *J. Chem. Phys.*, vol. 21, no. 6, pp. 1087–1092, Jun. 1953.
- [52] W. K. Hastings, "Monte Carlo sampling methods using Markov chains and their applications," *Biometrika*, vol. 57, no. 1, p. 97, 1970. [Online]. Available: https://academic.oup.com/biomet/article/57/1/ 97/284580
- [53] K. Saxena and A. R. Abhyankar, "Agent based bilateral transactive market for emerging distribution system considering imbalances," *Sustain. Energy, Grids Netw.*, vol. 18, Jun. 2019, Art. no. 100203, doi: 10.1016/j.segan.2019.100203.
- [54] E. Reihani, P. Siano, and M. Genova, "A new method for peer-to-peer energy exchange in distribution grids," *Energies*, vol. 13, no. 4, p. 799, Feb. 2020.

- [55] M. Babar, J. Grela, A. Ożadowicz, P. Nguyen, Z. Hanzelka, and I. Kamphuis, "Energy flexometer: Transactive energy-based Internet of Things technology," *Energies*, vol. 11, no. 3, p. 568, Mar. 2018.
- [56] Ofgem. (2020). Breakdown of an Electricity Bill. [Online]. Available: https://www.ofgem.gov.uk/data-portal/breakdown-electricity-bill
- [57] N. L. Johnson, "Systems of frequency curves generated by methods of translation," *Biometrika*, vol. 36, nos. 1–2, p. 149, Jun. 1949.
- [58] N. L. Johnson, "Bivariate distributions based on simple translation systems," *Biometrika*, vol. 36, nos. 3–4, p. 297, Dec. 1949.
- [59] T. Capper, J. Kuriakose, and M. Sharmina, "Monte Carlo simulation of the profits of households in a peer-to-peer electricity market accounting for imbalance charges," 2022.



JAISE KURIAKOSE is an expert in greenhouse gas accounting and energy system modeling. He has a strong track record of working at the boundaries of research and practice. He collaborated with various local/regional governments in the U.K. to develop carbon budgets and targets that align with the Paris Agreement; five U.K. cities/regions have formally adopted targets based on his work which has influenced a further 250 local authorities. Throughout his career, he has engaged with

diverse stakeholders from industry, NGOs, public, and academia. He was selected as a top 100 BAME climate expert in the U.K. by Climate Reframe, in 2020.



MARIA SHARMINA received the Ph.D. degree from the Tyndall Centre. Her educational background is in economics and statistics. She was a Visiting Fellow with the University College London Energy Institute. She is currently a Reader in energy and sustainability at the Tyndall Centre for Climate Change Research, School of Engineering. She is also a Senior Academic Advisor with the Government Office for Science and Department for Business, Energy and Industrial Strategy

(BEIS), working on the Net Zero Foresight Project. She is a member of the Programme Advisory Board for the U.K. Research and Innovation (UKRI) Digital Economy Theme and the Sustainable Robotics Committee at the British Standards Institution (BSI). She is the University of Manchester Principal Investigator on the UKRI £8m Energy Revolution Consortium, leading the work on innovative business models for the energy sector.

...



TIMOTHY CAPPER (Member, IEEE) received an M.Eng. degree in civil engineering from The University of Manchester, U.K., in 2012, where he is currently pursuing the Ph.D. degree with the Tyndall Centre for Climate Change Research. From 2012 to 2018, he worked with BP. His research interests include the design of electricity markets to allow greater participation of small distribution connected generation, storage and flexible loads, and the interactions between local energy markets and traditional energy markets.